Author's response to reviews

Title: Treatment of isoniazid-resistant pulmonary tuberculosis

Authors:

Yee Hyung Kim (yeehyung0@paran.com)
Gee Young Suh (suhgy@skku.edu)
Man Pyo Chung (mpchung@skku.edu)
Hojoong Kim (hik3425@skku.edu)
O Jung Kwon (ojkwon@skku.edu)
Seong Yong Lim (symd21.lim@samsung.com)
Si Young Lim (siyoung.lim@samsung.com)
Won-Jung Koh (wjkoh@skku.edu)

Version: 7 Date: 16 January 2008

Author's response to reviews:

January 16, 2008

Dear Editor:

We revised our previous manuscript entitled "Treatment of isoniazid-resistant pulmonary tuberculosis (Manuscript ID 1969953558126524)" following the requests from the BioMed Central Editorial Production Team.

1. Section headings: Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Abstract not ABSTRACT

Correction was made as was recommended.

2. References - The reference list should contain all authors names, if there are 30 or below. The term et al. should not be used.

Correction of the reference 3 was made as was recommended.

We hope that our reply and all changes we have made meet with your approval and look forward to your response.

Sincerely,

Won-Jung Koh, MD

Division of Pulmonary and Critical Care Medicine
Department of Medicine
Samsung Medical Center
Seoul, Korea